1

Scientific Impact Paper No. XX

Peer review_draft – March 2024

Care of Women with Preterm Prelabour Rupture of the Membranes Prior to 24 Weeks' Gestation

2

1

1. Scientific Impact Paper

Hall M, Care A, Goodfellow L, Milan A, Curran C, Simpson N, Heazell A, Quenby S, David A, Shennan A, Story L, on behalf of the Royal College of Obstetricians and Gynaecologists

Correspondence: Royal College of Obstetricians and Gynaecologists, 10-18 Union Street, London SE1 1SZ

4

3

5 Plain language Summary

- 6 Rupture of the membranes is commonly referred to as 'waters breaking'. This usually occurs just before
- 7 or during labour. In around three in 100 pregnancies it occurs before 37 weeks of pregnancy (preterm),
- 8 but the woman does not labour within 24 hours: this is preterm prelabour rupture of the membranes
- 9 (PPROM). These women often give birth preterm. This paper looks at PPROM before 24 weeks of
- 10 pregnancy. This happens in a much smaller number of women.
- 11 PPROM prior to 24 weeks is particularly concerning because of the chance of the baby being born
- 12 extremely preterm. Babies born before 22_weeks cannot survive. Babies born between 22 and 26 weeks
- 13 are at risk of severe and sometimes life-long problems. They also have a higher risk of dying than babies
- 14 born later. Women sometimes develop an infection after PPROM, which can be extremely dangerous. If
- 15 this happens, doctors will suggest ending the pregnancy even if the baby would not survive so that the
- 16 woman does not become unwell (termination for a medical reason). However, some babies do survive
- 17 and are discharged home, well, and most mothers have no long-term physical problems.
- 18 This situation is very difficult for women and people who are pregnant. It is made more so by a lack of
- 19 clear information for doctors and midwives about how well women and babies in this situation will do,
- 20 and how to look after them. This results in lots of variation in information and care for women.
- 21 Here we summarise the current evidence about this condition. Firstly, we explain available information
- 22 on how well women and babies are likely to do. Then we discuss evidence about predicting what
- 23 problems individual women and babies might have. Finally, we look at evidence on the ways in which we
- 24 can care for women and their babies up to delivery.
- 25

This guidance is for healthcare professionals who care for women, non-binary and trans people who experience PPROM. Within this document we use the terms woman and women's health. However, it is important to acknowledge that it is not only women for whom it is necessary to access women's health and reproductive services in order to maintain their gynaecological health and reproductive wellbeing. Gynaecological and obstetric services and delivery of care must therefore be appropriate, inclusive and sensitive to the needs of those individuals whose gender identity does not align with the sex they were assigned at birth.

34 1. Introduction

35 Preterm prelabour rupture of the membranes (PPROM) occurs when the fetal membranes rupture prior 36 to 37 weeks of gestation and is associated with a variety of adverse maternal and fetal outcomes. Risk of 37 mortality and severe morbidity is inversely associated with gestational age at membrane rupture. While 38 there is a growing body of evidence on management of PPROM at or after 24 weeks' gestation, which has resulted in recent comprehensive clinical guidance,^{1, 2} there is a paucity of evidence and guidance 39 regarding optimal management of PPROM prior to this. PPROM at less than 24 weeks gestation occurs in 40 41 at least 1 in 2750 pregnancies and represents a group at particularly high risk of maternal and perinatal morbidity and mortality.³ In addition to the complexities surrounding management of pregnancies at risk 42 43 of imminent delivery at the extremes of gestational age and birthweight,⁴ PPROM at this gestation may be a clinical indication for termination of pregnancy for medical indication (TFMR).² Uncertainty 44 surrounding clinical outcomes as well as complex management decisions also leaves women at high risk 45 of psychological morbidity.^{5, 6} There is no national guidance on this condition, and women who have 46 47 experienced it describe significant variations in counselling and practice. Regarding language used in this 48 guideline, previable delivery has a variable definition internationally based, but is taken to mean delivery prior to 22 completed gestational weeks. The evidence in this SIP relates to spontaneous PPROM; while 49 50 there is likely to be significant overlap with iatrogenic PPROM, professionals should be wary of applying the information stated here to this different clinical scenario. 51

52 The purpose of this Scientific Impact Paper is to advise on emerging evidence on the outcomes and 53 management of PPROM at <24 weeks' gestation, and its implications for practice and future research. 54 This is achieved via review of published literature and international guidance, and with collaboration from 55 relevant patient groups. Commented [MH1]: UKOSS https://www.medrxiv.org/content/10.1101/2023.03.07.232 86863v1

Commented [MH2]: Ref Fiona Challacombe once published
Commented [MH3]: Ref Fiona Challacombe once published

56 2. Outcomes of pregnancies following PPROM <24 weeks' gestation

57 Counselling of women with PPROM and their families is critical to facilitate informed decision-making. 58 Historical concerns about invariably poor fetal and neonatal outcomes are changing as neonatal care 59 advances, and women should be counselled based on individual risk, including gestation at membrane 60 rupture. Nonetheless, the risk of maternal deterioration, fetal demise, previable birth, or at the extremes 61 of viability with the associated risks of neonatal death or long-term neurodisability, renders counselling 62 complex - particularly concerning decisions for termination of pregnancy versus expectant management.

Of note, all research on PPROM is limited by diagnostic techniques utilised. The gold standard remains 63 visualisation of amniotic fluid in the posterior fornix.¹ Given the smaller volume of amniotic fluid at lower 64 65 gestations, it is possible this would be harder to identify prior to 24 weeks. It has become clinical practice 66 to use any of a number of commercially available bedside immunochromatographic tests where the 67 diagnosis is equivocal.⁷ However, evidence for these tests is lacking, with initial investigation often 68 undertaken against now discredited tests, such as ferning or nitrazine testing. Even taking this into account, false positive rates of up to 9% have been suggested (likely an underestimation given the 69 70 validation techniques described above).⁸ Furthermore, study conditions where women have 'signs and 71 symptoms' are not replicated in clinical practice where tests are used in equivocal cases (where women 72 tend to have symptoms but not signs).⁹ Ultrasound is not recommended as a diagnostic tool for PPROM 73 owing to a lack of sensitivity; where oligohydramnios is noted and the diagnosis of PPROM is equivocal, 74 then fetal medicine specialist review would be warranted to rule out other causes of anhydramnios, such 75 as a severe renal anomaly. As well as potentially poor diagnostic accuracy being a consideration when 76 reviewing research, improved diagnostic techniques is an essential aspect of research going forwards as 77 well as a limitation of our clinical abilities that women should be made aware of during counselling.

78 2.1 Maternal outcomes

79 The UK Obstetric Surveillance System (UKOSS) study on Preterm Prelabour Rupture of the Membranes 80 prior to 23 weeks' gestation prospectively collected data nationally from women with pregnancies complicated by PPROM prior to 23 weeks' gestation between 1st September 2019 and 28th February 2021 81 82 (a subgroup analysis was undertaken to consider the impact of the Covid-19 pandemic and no difference 83 was seen, so the entire dataset is considered here). This demonstrated a 10% maternal sepsis rate among 84 women who had TFMR, compared to 13% among women who initially had expectant management. The 85 maternal mortality rate reported was 0.5% (~55/10,000 women, both secondary to sepsis), although a 15-86 year analysis of the French Confidential Enquiries into Maternal Deaths gave a more conservative estimate

Commented [MH4]: Ref UKOSS

with a previable PPROM attributable maternal mortality rate of 0.6/10,000, and a mortality rate among women with previable PPROM of 4.5/10,000 (95% CI 1.4-9.2) (previable defined as 14⁺⁰ to 24⁺⁶ gestational weeks).¹⁰ Whether PPROM occurs secondary to chorioamnionitis, or chorioamnionitis occurs secondary to the loss of the maternal-fetal barrier after PPROM is unclear and likely dependent on underlying pathology; in any case, genital tract sepsis remains a common cause of death. In the UK Confidential Enquiries into Maternal Deaths; the 2019-2021 report highlighted that two (of a total of 241 maternal deaths in the triennia) occurred following sepsis directly attributable to second trimester PPROM.¹¹

94 A retrospective study from three institutions in the USA studied 208 women in three US institutions who 95 experienced PPROM before 24 weeks' gestation between 2011-2018 and who either chose expectant 96 management (51.9%) or TFMR (48.1%). Compared to women who chose TFMR, women who chose 97 expectant management had 4.1 times increased risk of developing chorioamnionitis (38.0% vs 13.0%; 95% 98 confidence interval, 2.03-8.26, p<0.001) and 2.44 times the odds of postpartum haemorrhage (23.1% vs 99 11.0%; 95% confidence interval, 1.13-5.26, p=0.027). Admissions to the intensive care unit and unplanned hysterectomy only occurred after expectant management (2.8% vs 0.0 and 0.9% vs 0.0 respectively). Of 100 101 women who chose expectant management, 36.2% delivered via Caesarean section with 56.4% not having 102 a low transverse incision to the uterus. Composite maternal morbidity rates (encompassing 103 chorioamnionitis, unplanned surgery, unplanned hysterectomy, blood product transfusion and intensive care unit admission) were 60.2% in the expectant management group and 33.0% in the TFMR group 104 (p<0.001). After adjusting for gestational age at PPROM, site, race and ethnicity, gestational age at entry 105 to prenatal care, PPROM in a previous pregnancy, twin pregnancy, smoking, cervical cerclage, and cervical 106 107 examination at the time of presentation, expectant management was associated with 3.47 times 108 increased risk of composite maternal morbidity (95% confidence interval, 1.52-7.93), corresponding to an adjusted relative risk of 1.91 (95% confidence interval, 1.35-2.73). Among women who chose expectant 109 110 management, 15.7% avoided morbidity and had a neonate who survived to discharge.¹² While this study 111 did not comment on the need for manual removal of placenta specifically, the UKOSS study also 112 highlighted a 20% rate among all women with PPROM prior to 23 weeks' gestation regardless of 113 gestational age at delivery, which is in line with previous reports.³

114 2.2 Fetal and neonatal outcomes

115 Major risks to the survival of a fetus and neonate following PPROM include previable birth, complications 116 of extreme prematurity, pulmonary hypoplasia, overwhelming sepsis, and other PPROM-associated 117 complications such as cord prolapse or placental abruption. While the British Association of Perinatal

118 Medicine (BAPM) framework provides useful data on neonatal survival for counselling parents when birth is imminent at extremes of gestational ages, it does not mention PPROM or chorioamnionitis as non-119 120 modifiable risk factors that should be used to adjust the risk of a poor outcome,⁴ thereby limiting its 121 usefulness in this group. Latency to delivery is influenced by gestational age at PPROM (Table 1), a factor 122 that must be considered in counselling, both in terms of risk of second trimester pregnancy loss but also 123 when considering best place of care (i.e., whether admission and/or transfer of care and planning for 124 delivery in a tertiary unit is most appropriate). Data described in Table 1 includes all expectantly managed 125 cases and it must be noted that there was a 56% intrauterine death or previable delivery rate in this group, including a 16% intrauterine death rate among cases delivered from 22^{+0} weeks' gestation onwards. 126

	Gestational weeks at preterm prelabour rupture of the membranes, n (%)					
Latency to delivery	16+0 - 17+6	18+0 - 19+6	20+0 - 21+6	22+0 - 22+6		
	n=43	n=70	n=80	n=30		
<72 hours	16 (37 <u>)</u>	18 (26)	20 (25 <u>)</u>	6 (20 <u>)</u>		
72 hours to <7_days	27 (9 <u>)</u>	8 (11 <u>)</u>	9 (11 <u>)</u>	6 (20 <u>)</u>		
7 days to <28 days	6 (14 <u>)</u>	12 (17 <u>)</u>	24 (30 <u>)</u>	6 (20 <u>)</u>		
<u>></u> 28 days	17 (40 <u>)</u>	32 (46 <u>)</u>	26 (33 <u>)</u>	10 (33 <u>)</u>		
<u>Unspecified</u>	0	0	1 (1)	2 (7)		

Table 1: Latency to delivery by gestational age at preterm prelabour rupture of the membranes following decision for expectant management (data from the UKOSS study, including spontaneous onset and induction of labour, but excluding termination of pregnancy).

The UKOSS study findings highlighted that 31% of women elected to terminate (with the highest rate of 130 131 termination seen in women who had PPROM at under 18 weeks' gestation) and 69% elected to continue 132 with the pregnancy. Of women continuing with a singleton pregnancy, 44% (98/223) had a liveborn child, 133 and 18% (38/207) had a child that survived to hospital discharge without severe morbidity. Severe 134 morbidity was defined as grade 3 or 4 intraventricular haemorrhage and/or requirement for oxygen at 36 135 weeks postmenstrual age (the commonly used definition of bronchopulmonary dysplasia). The range of 136 worst-best outcomes if women who had a termination were included within the analysis are a livebirth 137 rate of 30-62% and a child survival to discharge without severe morbidity of 12-48%. Of note, there was 138 no significant difference between morbidity outcomes of survivors of earlier versus later gestation at 139 PPROM (Table 2), although there was a trend towards higher rates of survival to discharge without severe 140 morbidity if PPROM occurred from 20 weeks' gestation. Longer term outcomes are not available.

Commented [MH5]: Ref UKOSS

Deleted:

6

	Gestational weeks at preterm prelabour rupture of the membranes, n(%)				
Outcome	16+0 - 17+6	18+0 - 19+6	20+0 - 21+6	22+0 - 22+6	р
	n=84	n=102	n=107	n=37	
Livebirth	14 (33 <u>)</u>	27 (39 <u>)</u>	37 (4 <u>6)</u>	20 (67 <u>)</u>	0.023
Survival to discharge	17 (8 <u>)</u>	16 (24 <u>)</u>	21 (28 <u>)</u>	10 (38 <u>)</u>	0.265
Discharge without severe morbidity	5 (13 <u>)</u>	11 (16 <u>)</u>	13 (59 <u>)</u>	9 (35 <u>)</u>	0.127
Termination for medical reasons	39 (46)	32 (31)	25 (23)	7 (19)	0.004

Table 2: Neonatal outcome following preterm prelabour rupture of the membranes. (data from the UKOSS
 study).

Commented [MH6]: Ref UKOSS

Severe morbidity is defined as: grade 3-4 intraventricular haemorrhage or supplemental oxygen requirement at or beyond 36 weeks' postmenstrual age.

An older retrospective single-centre study from The Republic of Ireland identified 42 women with PPROM 146 147 before 24 weeks of gestation between 2007 and 2012 (when termination was not possible until the 148 woman's life was in danger) indicated a livebirth rate of 24% but with only 5% of infants surviving to discharge.³ Of note, mean gestation at membrane rupture was 18 weeks and delivery 20+5 weeks' 149 gestation, as compared to 19+3 and 22+4 weeks' gestation (for women not having termination) 150 respectively in the UKOSS study, which provides some explanation for the discrepancy in infant survival 151 152 data. Although it is equally plausible that this difference is at least partially attributable to improved neonatal care over time for extremely preterm babies. There is a growing body of evidence looking at 153 neonatal outcomes of survivors. The EPIPAGE-2 study conducted a secondary analysis of outcomes in 154 155 PPROM from 22-25 weeks' gestation and demonstrated a 10.5% and 36.0% survival to 2 years without cerebral palsy in babies where PPROM had occurred at 22 and 23 weeks' gestation respectively.¹³ Another 156 157 retrospective study (that excluded women undergoing termination of pregnancy) gives a 49% survival rate to discharge among neonates following PPROM at 20-24 weeks' gestation, with 47% of survivors 158 experiencing severe neonatal morbidity; the mortality rate after discharge from neonatal care was not 159 recorded.¹⁴ One study compared outcomes of early (<25 weeks) and later (25-31 week) PPROM 160 161 demonstrating a significantly higher rate of severe morbidity (51.5 vs 22.5%; defined as moderate to 162 severe cerebral palsy or a Bayley II score more than two standard deviations below the mean) among 163 survivors in the early PPROM group.¹⁵ However, these neurological differences may represent the impact of chorioamnionitis on the preterm brain, rather than the impact of PPROM alone, with significantly 164 increased rates of per- and intraventricular haemorrhage, intracerebral haemorrhage and neonatal 165

seizures demonstrated in a study of 9,633 neonates born prior to 34 weeks' gestation with chorioamnionitis as compared to those without.¹⁶

168 2.3 Prediction of outcomes

169 2.3.1 Prediction of pulmonary hypoplasia

170 Amniotic fluid is vital in antenatal lung development, both in terms of achieving normal volume and production of important mediators of subsequent pulmonary function such as surfactant. Pulmonary 171 hypoplasia, defined as a reduction in lung cells, airway, alveoli resulting in reduced organ size, but 172 173 practically almost always used to refer to a reduction in alveoli, can occur secondary to PPROM with 174 incidence increasing with decreasing gestational age.¹⁷ Given that formal diagnosis requires post-mortem 175 assessment, postnatal identification is also challenging and largely based on secondary complications such as pulmonary hypertension or high oxygen requirements.¹⁸ One systematic review of outcomes following 176 PPROM prior to 24 weeks' gestation identified one study that looked specifically at survival of babies with 177 clinical pulmonary hypoplasia, quoting a 64% mortality rate in affected liveborn infants (mean latency to 178 179 delivery 20-43 days). Although they note that this figure may under-represent true mortality as babies who died in the first 24 hours of life were less likely to have a clinical diagnosis prior to death.¹⁹ 180 181 Amniotic fluid assessment has been investigated to determine risk of pulmonary hypoplasia. A prospective

study of 580 women with PPROM between 20 and 28+0 weeks' gestation has demonstrated that a single deepest vertical pool (SDVP) of <2cm at presentation is associated with worse respiratory outcomes.²⁰ A smaller study of 31 women has suggested reduced neonatal survival where the SDVP is <1cm.²¹ Both studies suggest that a higher SDVP increases latency to delivery, which could explain the improvement of respiratory and survival outcomes independent of the SDVP.

187 Two-dimensional ultrasound measures, such as the thoracic circumference, lung:head ratio²² and quantitative lung index (= lung area/(head circumference/10)²) have been evaluated as prognostic 188 189 markers for pulmonary hypoplasia and poor outcome in fetuses with congenital diaphragmatic hernia.^{23,} ²⁴ However, these techniques are not validated in women with second trimester PPROM, and many 190 191 studies are limited by verification bias of diagnosis of membrane rupture; therefore, these techniques have limited prognostic accuracy.²⁵ Three-dimensional ultrasound using virtual organ computer aided 192 193 analysis, has been demonstrated to have good prediction of lung volumes in pulmonary hypoplasia as 194 compared to post-mortem volumes. However, it is technically challenging, and is severely limited by fetal 195 position and acoustic shadow, including from a lack of amniotic fluid, so is not clinically useful. While 196 multiplanar 3D ultrasound is less technically challenging, its results have not been shown to predict

neonatal outcome.²⁶ Although magnetic resonance imaging (MRI) is not validated for pulmonary hypoplasia prediction it may carry value in overcoming sonographic challenges associated with anhydramnios; one small study has demonstrated that volumetry can be used to predict neonatal mortality secondary to respiratory distress following PPROM between 16 and 27 weeks' gestation.²⁷ A larger trial using MRI to predict pulmonary hypoplasia is underway. Despite progress in other congenital pulmonary conditions, the difficulty in prediction of pulmonary hypoplasia in PPROM limits individual counselling and neonatal planning.

204 2.3.2 Prediction of arthrogryposis

Arthrogryposis, multiple congenital limb contractures, is a condition with heterogenous aetiology 205 sometimes associated with PPROM at <24 weeks' gestation owing to reduced potential for fetal 206 207 movements.²⁸ The prevalence of arthrogryposis associated with PPROM is not well documented: the 208 UKOSS study reports two cases of 54 survivors with one or two limbs affected; a retrospective study of 209 130 neonates born following PPROM prior to 24 weeks' gestation describes a 29% rate of limb contractures;¹⁴ larger studies of the aetiology of arthrogryposis demonstrates a much lower incidence 210 suggesting the prevalence of arthrogryposis 1/3000 overall, with only around 1% of these cases associated 211 212 with any cause of oligohydramnios.²⁹

The rarity of arthrogryposis and its diverse aetiology results in available evidence being difficult to interpret in the context of PPROM. Prediction of arthrogryposis is challenging, with around 75% of cases not diagnosed in the antenatal period (all aetiologies).³⁰ Arthrogryposis in the context of PPROM is likely to be even more difficult to diagnose owing to poorer quality imaging in the presence of oligohydramnios, and an absence of other syndromic findings pointing towards a diagnosis. Treatment is widely varied; arthrogryposis secondary to oligohydramnios is normally responsive to physical therapies,²⁹ whereas syndromic causes are more commonly associated with a need for surgeries.³¹

220 2.3.3 Prediction of maternal and fetal infection

While expedition of delivery in cases of clinical chorioamnionitis is essential in providing safe obstetric care, reliable antenatal diagnosis of infection remains elusive. Current practice of monitoring maternal symptoms, white cell count (WCC) and C-reactive protein (CRP) are of limited value across all gestations. While maternal pyrexia is sensitive (94-100%) at temperatures at and above 38°C, it is non-specific in the absence of other symptoms, most of which are relatively insensitive (for example, maternal tachycardia 50-70% sensitive; foul smelling discharge 5-22% sensitive).³² While there may be some concern regarding method of monitoring temperature, studies in adults have demonstrated axillary assessment with a digital Commented [MH7]: Ref UKOSS when available

thermometer most reliable, excluding the significantly more time-consuming 12-minute gallium in glass test, which was most reliable overall.³³ Significantly, there are inconsistencies throughout the literature on the definition of a pyrexia, with a range from 37.5-38.3°C reported by studies, which hampers interpretation of predictive value, with some also including temperatures of below 36°C within their analysis;^{32, 34-36} therefore, it is noteworthy that current UK guidance on determining the presence of clinical chorioamnionitis in PPROM does not define a threshold for pyrexia.^{1, 7}

234 Likewise, WCC is relatively sensitive in the presence of corroborating symptoms, but not useful without 235 them; CRP has not been demonstrated to be of value.³² There has been some interest in the neutrophil to 236 lymphocyte ratio as a marker of chorioamnionitis in clinically well women, but this has not been studied 237 specifically in the context of PPROM at any gestation, nor have there been attempts to analyse the impact 238 of integration into clinical practice.³⁷ No study has examined these parameters in early gestations 239 specifically, although there is no reason to think they would be more reliable. While there is evidence that 240 an increase in fetal heart rate of greater than 10% from baseline is associated with term chorioamnionitis,³⁸ this has not been replicated in the preterm group. Fetal tachycardia is likely to be less 241 sensitive at early gestations owing to the physiological effects of unopposed sympathetic activity.³⁹ 242

Multiple studies have examined the intraamniotic environment via either amniocentesis or transvaginal collection of amniotic fluid following PPROM. Studies including women with PPROM prior to 24 weeks' gestation have suggested diagnostic utility of multiple markers including, but not limited to, interluekin- 40 matrix metalloproteinase-8, ⁴¹ monocyte chemoattractant protein-1 ⁴² and tumour necrosis factor- α .⁴³ In particular, interleukin-6 has been investigated, including on bedside immunochromatography, but not prior to 24 weeks' gestation.⁴³⁻⁴⁶ However, there is a paucity of large-scale clinical trials, and so no significant translation into clinical practice.

Fetal imaging to diagnose fetal inflammatory response is also an active area of research.⁴⁷ There have been multiple attempts to determine the value of ultrasound Dopplers in predicting a clinical diagnosis of chorioamnionitis: a retrospective study of 504 women with PPROM from 23 to 34 weeks' gestation compared those with and without a suspected chorioamnionitis and confirmed no difference in umbilical or middle cerebral artery pulsatility index, with a poor predictive value in both tests (area under the curve [AUC] 0.619, 95% CI 0.424-0.813 and AUC 0.442, 95% CI 0.265-0.618 respectively).⁴⁸ To our knowledge, no work undertaken at earlier gestational ages is available.

A meta-analysis of 12 studies of 1,744 participants found that chorioamnionitis is more common when ultrasound assessed thymic size is decreased (73.9% of cases compared to 27.1%), although none of the studies included pregnancies at less than 24 weeks' gestation.⁴⁹ While small studies have attempted to utilise assessment of adrenal glands to predict preterm birth, none have specifically attempted to determine the impact of chorioamnionitis.⁴⁷ Studies are ongoing looking at the utility of MRI given promising differences in predicting delivery in women at high risk of delivery prior to 32 weeks' gestation.⁵⁰⁻⁵²

264 Complementary to ongoing clinical studies are recent developments in animal models to aid 265 understanding of pathophysiology of chorioamnionitis with and without PPROM. Extensive ovine work 266 investigating the impact of lipopolysaccharide (LPS) induced chorioamnionitis on individual fetal organs is 267 likely to inform decisions on imaging targets.⁵³⁻⁵⁶ Furthermore, significant steps have been made to 268 address the longstanding concern regarding whether LPS can truly replicate clinical infection by 269 development of a murine model of intravaginal E. coli infection.⁵⁷ Ongoing close working between the 270 basic and clinical sciences remains key in improving knowledge and outcomes.

271 3. Antenatal management of pregnancies affected by PPROM <24 weeks' gestation

272 3.1 Place of care

273 One study has evaluated risks of outpatient management in women with PPROM at any gestation. Women with PPROM prior to 26 weeks' gestation were found to have a significantly increased risk of complications 274 275 (fetal or neonatal death, placental abruption, umbilical cord prolapse or delivery outside of a maternity unit) if managed as an outpatient (odds ratio [OR] 6.2, 95% confidence interval [CI] 1.6 - 23.8).58 Although 276 277 this does not negate the role of outpatient management, women should be considered high risk and there 278 should be a very low threshold for admission. Where there are clinical concerns about evolving sepsis or 279 impending abruption, women must remain as inpatients. When a decision has been made for 280 consideration of neonatal resuscitation at delivery, women should be cared for in a unit with suitable 281 neonatal facilities; in complex cases assessment by a fetal medicine specialist and a senior neonatologist 282 would allow for site-specific decision making.

283 3.2 Antibiotic use

Optimal gestation to commence a course of prophylactic oral antibiotics is unclear, as is choice of antibiotic and duration of course. Rationale for administration after 24 weeks' gestation is from a Cochrane review that demonstrates increased latency to delivery and reduction in short-term neonatal complications, without impact on maternal or neonatal mortality, or long-term infant outcomes when

antibiotics are given.⁵⁹ However, this study is all gestations, and there are no subgroup analyses by gestational age at PPROM; the numbers of women with PPROM prior to 24 weeks' gestation who are included is unclear.

The largest study to date of oral antibiotic use in PPROM recruited 4,826 women into a randomised placebo-controlled trial. While there was no lower gestational age for inclusion, there was no subgroup analysis for very early gestations. Administration of erythromycin rather than placebo was associated with a significantly increased latency to delivery of 48 hours, as well as a reduction in composite neonatal morbidity.^{60, 61}

3.3 Antenatal corticosteroid and magnesium sulphate use

Evidence for the use of antenatal corticosteroids (ACS) prior to 24 weeks' gestation is lacking. One observational study carried out over 15 years demonstrated a reduction in death or neurodevelopmental delay in babies born at 23 weeks' gestation or later (68.4% versus 90.5%); the same was not true of babies born at 22 weeks' gestation.⁶² There is no higher quality evidence than this. BAPM does not support universal use prior to 24 weeks' gestation.⁴

There is increasing evidence that administration of antenatal steroids close to time of delivery confers 302 greatest risk reduction; therefore, ACS should ideally not be given more than seven days prior to delivery, 303 and repeated doses avoided as they are associated with reduction in birthweight and may worsen 304 305 neurodevelopmental outcomes. Given the higher rate of pulmonary hypoplasia in neonates born 306 following PPROM at under 24 weeks' gestation, incorrect timing of steroids is likely to have a particularly 307 detrimental effect. There has been some concern regarding the administration of steroids to women at 308 high risk of infection. While the ACT trial, which was performed in seven low- and middle-income countries, did show a trend towards increased rates of chorioamnionitis among women who received ACS 309 (OR 1.46, 95% CI 0.81-2.66),63 this has not been replicated in the most recent Cochrane review, which 310 included 1 5 RCTs, including the ACT trial (OR 0.86, 95% CI 0.69-1.08).64 311

There is no evidence for the use of magnesium sulphate prior to 24 weeks' gestation, although it would be pragmatic to consider this if steroids have been given and there is a plan for neonatal resuscitation.

314 3.4 Bedrest

There is no evidence supporting the use of bedrest to improve outcomes of PPROM at any gestation: a pilot randomised control trial of 32 women with PPROM from 24 weeks' gestation demonstrated no maternal or neonatal benefit.⁶⁵ A single-centre study over a two-year period found a significantly

increased risk of venous thromboembolism (VTE) in women advised three or more days of bedrest as part of the management of PPROM as compared to the background population (15.6 cases per 1000 deliveries, and 0.8 per 1000 deliveries respectively) without any obstetric benefit.⁶⁶ However, it should be noted that national recommendations for VTE prophylaxis at the time of this study, would have resulted in no women being given low molecular weight heparin (LMWH).⁶⁷ Nonetheless, current guidance would not insist upon LMWH⁶⁸ and decision-making surrounding prescription is complicated by risk of labour, meaning that these results continue to have validity even if awareness around the risk of VTE is greater now.

325 3.5 Management of pregnancies with cerclage in situ

Absolute indications for the removal of a cervical cerclage are no different in women prior to 24 weeks'
 gestation and include: confirmed labour, ongoing antepartum haemorrhage, maternal sepsis, fetal
 demise, and decision for imminent vaginal delivery.⁶⁹

The best course of action for management of cerclage in women with PPROM prior to 24 weeks' gestation 329 330 and no absolute indication for delivery is uncertain. Existing evidence is limited in its application given 331 higher gestational ages at membrane rupture and variable antibiotic protocols. A recent systematic review 332 and meta-analysis of cerclage removal versus retention at all preterm gestations following PPROM 333 demonstrated a decreased risk of delivery withing 48 hours in the retention group (OR 0.15, 95% CI 0.07-334 0.31), but decreased rates of chorioamnionitis and one minute Apgar <7 in the removal group (OR 0.57, 95% CI 0.34-0.96 and OR 0.22, 95% CI 0.08-0.56 respectively).⁷⁰ Another review of multiple studies 335 proposed that cerclage retention is associated with increased rates of maternal pyrexia and 336 chorioamnionitis without improved latency.⁷¹ However, in all cases antibiotic use was not consistent 337 338 between studies, and poor outcomes seem to be associated with no antibiotic use, especially given the 339 apparent better outcomes in more recent work (where antibiotic protocols are in place).⁷² One study 340 evaluated impact of cerclage retention or removal across gestations, and demonstrated a significantly 341 increased risk of chorioamnionitis in the cerclage retention group if PPROM occurred prior to 28 weeks' 342 gestation.⁷³ No group has demonstrated neonatal benefit following cerclage retention or removal at time 343 of PPROM.

344 **3.6 Investigation and management of group B streptococcus**

Current RCOG guidance on the management of group B streptococcus (GBS) in pregnancy does not recommend GBS testing after PPROM.⁷⁴ This is a pragmatic recommendation as current NICE guidance for intrapartum GBS prophylaxis is risk-based and all cases of preterm labour or women with ruptured

membranes for >24 hours would receive intrapartum antibiotics regardless of GBS status on swab..⁷ Furthermore, current methods for GBS testing (low vaginal and anorectal swab) are not validated in PPROM. Evidence for whether routine testing for GBS in women with PPROM impacts outcome is lacking at all gestations. While this would be prudent at all gestations, investigation at early gestation may yield valuable information on balancing the risks of GBS sepsis and serious prematurity-associated morbidity.

353 3.7 Termination of pregnancy

- The most recent UK data demonstrates that 31% of women with PPROM prior to 23 weeks' gestation
- 355 will elect to undergo termination, with the decision being more common when PPROM occurs at earlier
- 356 gestations, Grounds for termination include risk to maternal health (risks of expectant management
- 357 versus TFMR are discussed in 2.1), and concerns about perinatal morbidity and mortality (see section
- 358 2.3.1). In cases where TFMR is not because of risk to the woman's life, current RCOG advice on the use
- 359 of feticide prior to termination should be followed (generally, to be offered where termination is to
- 360 occur after 21⁺⁶ weeks' gestation). Where feticide is not being performed, women should be advised of
- 361 the risk of signs of life following delivery based on their gestation.⁷⁵

362 3.8 Amnioinfusion and amniopatching

- A Cochrane review evaluating amnioinfusion in the 3rd trimester for women with PPROM was undertaken in 2014, finding sparse data and lack of methodological robustness.⁷⁶ The AMNIPROM pilot study demonstrated feasibility of amnioinfusion studies, but highlighted longer term neonatal outcomes as necessary endpoints.⁷⁷ Trials are currently underway in Germany for the management of 2nd trimester PPROM with amnioinfusion.⁷⁸
- Amniopatching was considered in a 2016 Cochrane review. Two studies, both deemed at high risk of bias, were included and there was considered to be inadequate evidence for recommendation in clinical practice.⁷⁹ In any circumstance, neither procedure should be offered outside of a clinical trial.

371 3.9 Emotional support

Women with PPROM are at higher risk of antenatal anxiety, postnatal depression and posttraumatic stress disorder.^{5, 6, 80} Best therapies and management of this are not known despite the significant additional burden placed on women by these comorbidities. Furthermore, there is very limited evidence of the impact of PPROM and its complications on the partners of affected women. Women themselves describe well-informed medical teams, comprehensive information and compassionate care as necessary for improving their own feeling of psychological wellbeing, as well as more formal psychological support. Commented [MH8]: Ref UKOSS

Commented [MH9]: Ref Fiona Challacombe

378 3.10 Multiple pregnancies

Evidence of optimum management of, and outcomes related to second trimester PPROM in multiple pregnancies is lacking. Data from the UKOSS study (23 DCDA twins and 10 MCDA twins) demonstrated a 20% survival to discharge rate for both twins, with single twin survival in a further 17% of pregnancies. However, management was complicated in six cases by either single twin delivery or intrauterine demise prior to 22⁺⁰, highlighting the complexity in management of these cases. There is no evidence on relative outcomes when there is single twin PPROM with preserved amniotic fluid in the second twin.

385 4. Intrapartum management

386 4.1 Optimum timing of delivery

There is histopathological evidence from a single-centre that delaying delivery until 34 weeks' gestation in women with known genital tract GBS colonisation who have PPROM from 23 weeks' gestation is not associated with an increased rate of GBS chorioamnionitis. There is no subgroup analysis for early gestations.⁸¹ There is no equivalent evidence for women with known GBS carriage who have PPROM prior to 23 weeks' gestation.

Among women without GBS, the RCOG recommendation to delay delivery of women until 37 weeks' in the absence of an acute indication for delivery (e.g., suspicion of chorioamnionitis, abruption, cord prolapse) gestation is based on a Cochrane review that includes no women with PPROM prior to 28 weeks' gestation so the recommendation cannot be reliably extrapolated to this group.⁸² There is no evidence on optimising timing of delivery in women with PPROM prior to 24 weeks' gestation.

397 4.2 Mode of delivery

398 4.2.1 Prior to viability

There is no evidence on safety of medical versus surgical TFMR in cases of PPROM. It should be noted that, while chorioamnionitis may complicate surgical termination, it does not contraindicate it and may be the safest way to deliver for some women.⁸³ All women and people who are pregnant undergoing TFMR or having a second trimester pregnancy loss following PPROM should be offered treatment dose antibiotics for chorioamnionitis given the extremely high rate of postnatal diagnosis (94%).⁸⁴

404 4.2.2 At viability

There is currently limited evidence regarding optimal mode of delivery or use of intrapartum fetal monitoring in women labouring at periviable gestations. However, routine Caesarean section is not Commented [MH10]: Ref UKOSS.

407 recommended for the indication of periviable delivery alone as it has not been shown to decrease 408 mortality or intraventricular haemorrhage.⁸⁵ Of note, no analysis was carried out considering the 409 implications of PPROM on complexity of delivery at Caesarean section or vaginal birth of extremely 410 preterm infants, which would be of interest given that the absence of the amniotic sac may increase both 411 the risk of bony injury at attempts to deliver vaginally or at Caesarean, and also laceration to the fetus at 412 uterine entry during Caesarean section.

413 Evidence concerning the management of preterm labour with breech presentation is lacking. A 414 retrospective study of 86 women delivering between 26 and 29⁺⁶ days gestation revealed that planned 415 Caesarean section was associated with fewer 5 minute Apgar scores of <7, but no difference in neonatal mortality or major morbidity.⁸⁶ The same study demonstrated no statistically significant difference in the 416 417 rates of head entrapment by mode of delivery (13% and 6% for vaginal and Caesarean respectively). The 418 rate of neonatal death in cases where deliveries had been complicated by head entrapment trended 419 towards significance (4.8% and 0 for vaginal and Caesarean delivery respectively),⁸⁶ perhaps reflective of the surgical difficulty of lateral cervical incisions versus inverted T incision. Similarly to the above study, 420 421 no analysis was carried out taking the impact of PPROM into account. Current RCOG recommendations to 422 avoid routine amniotomy to reduce the risk of head entrapment, and lateral cervical incisions to relieve it 423 should be followed at all viable gestations.⁸⁷ No studies focus on management of fetuses in transverse lie, although women must be counselled that (unlike at higher gestation) this is not an absolute indication for 424 Caesarean section at periviability, and vaginal delivery is achievable. 425

426 Regarding longer term maternal risk following periviability Caesarean section there is an increased risk

427 of uterine rupture regardless of direction of uterine incision,⁸⁸ and case series evidence suggests that

428 this risk may be increased further if the woman then has a transabdominal cerclage. A summary of

429 advice regarding neonatal care is given in Appendix 2 for information.

430 4.3 Placental histopathology

In line with national guidance, the placenta must be sent to histopathology in all cases where PPROM has occurred prior to 24 weeks' gestation and delivery occurs before 32 weeks' gestation, and gross and macroscopic analysis should be undertaken.⁸⁹ The histopathological findings associated with chorioamnionitis are given in Table 3. A placental swab sent for microscopy, sensitivity and cultures may aid in decisions surrounding antibiotics, particularly where there is no response to broad spectrum antibiotics, but it should be noted that a positive swab does not confer a diagnosis of histological Commented [MH11]: Ref Masa Zdravovic when published

chorioamnionitis (with positive swabs being a more common finding).⁹⁰ Diagnosis is clinically useful in 437 maternal and neonatal sepsis, and can inform care in future pregnancies. While rates of chorioamnionitis 438 439 following PPROM at all gestations are thought to be in the region of 17-58%,⁹¹ this rises to 94% in pregnancies delivering between 21 and 24 weeks.⁸⁴ Vascular lesions, such as subchorionic haematomas, 440 441 are also more common in PPROM, and are inversely related to the presence of funisitis, suggestive of an alternative aetiology in some women.92 There is no research comparing management of future 442 pregnancies depending on identified placental lesions specifically following PPROM. However, two studies 443 444 of fetal deaths (one from the UK and another from the Netherlands) found that chorioamnionitis may 445 recur in subsequent pregnancies.^{93, 94}

Maternal Inflammatory Response			
Stage 1: acute subchorionitis or chorionitis	Grade 1: not severe		
Stage 2: acute chorioamnionitis –	Grade 2: severe – confluent polymorphonuclear		
polymorphonuclear leukocytes extend into	leukocytes or subchorionic microabscesses		
fibrous chorion and/or amnion			
Stage 3: necrotising chorioamnionitis –			
karyorrhexis of polymorphonuclear leukocytes,			
amniocyte necrosis, and/or amnion basement			
membrane hypereosinophilia			
Fetal Inflamma	atory Response		
Stage 1: chorionic vasculitis or umbilical phlebitis	Grade 1: not severe		
Stage 2: involvement of the umbilical vein and	Grade 2: severe – near-confluent intramural		
one or more umbilical arteries	polymorphonuclear leukocytes with attenuation		
	of vascular smooth muscle		
Stage 3: necrotising funisitis			

446 Table 3: Histopathological findings indicative of chorioamnionitis as per the Amsterdam Criteria.⁹⁵

447 5. Cost implications

While separate data on delivery following PPROM is not available, the financial cost of preterm birth is significant, both in terms of immediate neonatal care and lifelong support for resulting morbidities including learning support. UK figures, based on cost estimates from 2006, suggest an annual cost of £2.9bn related to preterm birth.⁹⁶ More recent data from Australia, suggests that the cost of schooling is around £40,000 and £3700 more per year for extreme and late preterm birth respectively, as compared

453	to term deliveries ⁹⁷ In any instance, increasingly sophisticated neonatal care is likely to result in increased			
454	short and long term costs associated with preterm birth.			
455				
456	5. Opinion			
457	• There is a lack of high-quality evidence regarding maternal and fetal outcomes following PPROM			
458	prior to 24 weeks' gestation; this results in poorer counselling of women which is highlighted by			
459	the variation in advice and care that women describe was offered to them. However, healthcare			
460	professionals can reduce this heterogeneity by insuring most up-to-date evidence is given to			
461	patients, and being cautious when using existing tools that do not include PPROM or			
462	chorioamnionitis in their modelling of counselling. Regardless of available evidence, counselling			
463	must always be compassionate and have women and their families at its core. Nonetheless,			
464	prediction of both maternal and perinatal outcome warrants high-quality investigation if			
465	counselling is to improve.			
466	There is minimal data on longer-term neonatal outcome, and no data on outcomes later in			
467	infancy and childhood. Prospective, longitudinal data collection should be undertaken.			
468	Appropriate timing of interventions routinely offered when PPROM occurs at a viable gestation			
469	and weight (for example, ACS and prophylactic antibiotics) are unclear and require high-quality,			
470	adequately powered research.			
471	• PPROM prior to 24 weeks' gestation does carry a maternal mortality risk, and women who			
472	choose expectant management must be adequately counselled on symptoms of sepsis, the need			
473	for early presentation and the likely clinical plan if there were concerns about maternal sepsis.			
474	Regardless of outcome, PPROM carries a risk of poorer maternal mental health outcomes. The			
475	timing and type of intervention that best mitigates this must be studied, and be prioritised for			
476	translation into clinical practice once results are available.			
477	• An eventual aim of all this research must be co-ordinated care, nationally and internationally,			
478	based on national guidance developed with relevant stakeholders and improved by high-quality			
479	training of relevant healthcare professionals.			
480				

482 References

Thomson AJ, on behalf of the Royal College of Obstetricians and Gynaecologists. Care of women 483 1. presenting with suspected preterm prelabour rupture of membranes from 24+0 weeks gestation. BJOG 484 485 2019: 126e 152-66. 486 2. American College of Obstetrics and Gynaecology. Prelabour rupture of membranes: ACOG Practice Bulletin Number 217. Obstet Gynecol. 2020;135:80-97. 487 Linehan LA, Walsh J, Morris A, Kenny L, O'Donoghue K, Dempsey E, et al. Neonatal and maternal 488 3. 489 outcomes following midtrimester preterm premature rupture of the membranes: a retrospective cohort study. BMC Pregnancy Childbirth. 2016;16:25. 490 491 4. British Association of Perinatal Medicine. Perinatal management of extreme preterm birth 492 before 27 weeks' gestation. A BPAM Framework for Practice. BPAM: 2019. 493 5. Zemtsov GE, Avram CM, Darling A, Dillon J, Wheeler S, Dotters-Katz SK. Incidence and Risk 494 Factors for Postpartum Depression among Women with Preterm Prelabor Rupture of Membranes. Am J 495 Perinatol. 2022;39(8):797-802. 496 6. Fairbrother N, Young AH, Zhang A, Janssen P, Antony MM. The prevalence and incidence of perinatal anxiety disorders among women experiencing a medically complicated pregnancy. Arch 497 498 Womens Ment Health. 2017;20(2):311-9. 499 National Institute for Health and Care Excellence. Preterm Labour and Birth. NICE clinical 7. 500 guideline 25 [NG25]. NICE: 2015. McQuivey RW, Block JE. ROM Plus(*): accurate point-of-care detection of ruptured fetal 501 8. 502 membranes. Med Devices (Auckl). 2016;9:69-74. 503 Thomasino T, Levi C, Draper M, Neubert AG. Diagnosing rupture of membranes using 9. combination monoclonal/polyclonal immunologic protein detection. J Reprod Med. 2013;58(5-6):187-504 505 94 Abrahami Y, Saucedo M, Rigouzzo A, Deneux-Tharaux C, Azria E, ENCMM Group . Maternal 506 10. 507 mortality in women with pre-viable premature rupture of membranes: An analysis from the French confidential enquiry into maternal deaths. Acta Obstet Gynecol Scand. 2022;101(12):1395-402. 508 509 Knight M, Bunch K, A. F, Patel RM, Kotnis R, Kenyon S, et al. Saving Lives, Improving Mothers' 11. 510 Care. Lessons leared to inform maternity care from the UK and Ireland Confidential Enquiry into 511 Maternal Deaths and Morbidity 2019-21. MBRACE-UK; 2021 Sklar A, Sheeder J, Davis AR, Wilson C, Teal SB. Maternal morbidity after preterm premature 512 12. rupture of membranes at <24 weeks' gestation. Am J Obstet Gynecol. 2022;226(4):558.e1-.e11. 513 514 13. Lorthe E, Torchin H, Delorme P, Ancel PY, Marchand-Martin L, Foix-L'Hélias L, et al. Preterm 515 premature rupture of membranes at 22-25 weeks' gestation: perinatal and 2-year outcomes within a 516 national population-based study (EPIPAGE-2). Am J Obstet Gynecol. 2018;219(3):298.e1-.e14. 517 Kibel M, Asztalos E, Barrett J, Dunn MS, Tward C, Pittini A, et al. Outcomes of Pregnancies 14. 518 Complicated by Preterm Premature Rupture of Membranes Between 20 and 24 Weeks of Gestation. 519 Obstet Gynecol. 2016;128(2):313-20. 520 Manuck TA, Varner MW. Neonatal and early childhood outcomes following early vs later 15. 521 preterm premature rupture of membranes. Am J Obstet Gynecol. 2014;211(3):308.e1-6. Venkatesh KK, Jackson W, Hughes BL, Laughon MM, Thorp JM, Stamilio DM. Association of 522 16. 523 chorioamnionitis and its duration with neonatal morbidity and mortality. J Perinatol. 2019;39(5):673-82. 524 17. Lauria MR, Gonik B, Romero R. Pulmonary hypoplasia: pathogenesis, diagnosis, and antenatal prediction. Obstet Gynecol. 1995;86(3):466-75. 525

526 18. Dassios T. Critical functional lung volumes in neonatal intensive care: evidence and clinical

527 applications. Pediatric Research. 2023;94(1):82-8.

528 19. Sim WH, Araujo Júnior E, Da Silva Costa F, Sheehan PM. Maternal and neonatal outcomes 529 following expectant management of preterm prelabour rupture of membranes before viability. J Perinat 530 Med. 2017;45(1):29-44. 531 20. Weiner E, Barrett J, Zaltz A, Ram M, Aviram A, Kibel M, et al. Amniotic fluid volume at 532 presentation with early preterm prelabor rupture of membranes and association with severe neonatal 533 respiratory morbidity. Ultrasound Obstet Gynecol. 2019;54(6):767-73. 534 21. Storness-Bliss C, Metcalfe A, Simrose R, Wilson RD, Cooper SL. Correlation of residual amniotic 535 fluid and perinatal outcomes in periviable preterm premature rupture of membranes. J Obstet Gynaecol 536 Can. 2012;34(2):154-8. 537 Nimrod C, Davies D, Iwanicki S, Harder J, Persaud D, Nicholson S. Ultrasound prediction of 22. pulmonary hypoplasia. Obstet Gynecol. 1986;68(4):495-8. 538 Quintero RA, Quintero LF, Chmait R, Gómez Castro L, Korst LM, Fridman M, et al. The 539 23. 540 quantitative lung index (QLI): a gestational age-independent sonographic predictor of fetal lung growth. 541 Am J Obstet Gynecol. 2011;205(6):544.e1-8. 542 24. Vintzileos AM, Campbell WA, Rodis JF, Nochimson DJ, Pinette MG, Petrikovsky BM. Comparison 543 of six different ultrasonographic methods for predicting lethal fetal pulmonary hypoplasia. Am J Obstet 544 Gynecol. 1989;161(3):606-12. 545 van Teeffelen AS, Van Der Heijden J, Oei SG, Porath MM, Willekes C, Opmeer B, et al. Accuracy 25. of imaging parameters in the prediction of lethal pulmonary hypoplasia secondary to mid-trimester 546 547 prelabor rupture of fetal membranes: a systematic review and meta-analysis. Ultrasound Obstet 548 Gynecol. 2012;39(5):495-9. Avena-Zampieri CL, Hutter J, Rutherford M, Milan A, Hall M, Egloff A, et al. Assessment of the 549 26. 550 fetal lungs in utero. Am J Obstet Gynecol MFM. 2022;4(5):100693. 551 Messerschmidt A, Pataraia A, Helmer H, Kasprian G, Sauer A, Brugger PC, et al. Fetal MRI for 27. 552 prediction of neonatal mortality following preterm premature rupture of the fetal membranes. Pediatr Radiol. 2011:41(11):1416-20. 553 Hall JG. Arthrogryposis (multiple congenital contractures): diagnostic approach to etiology, 554 28. 555 classification, genetics, and general principles. Eur J Med Genet. 2014;57(8):464-72. 556 29. Hall JG. Oligohydramnios sequence revisited in relationship to arthrogryposis, with distinctive 557 skin changes. Am J Med Genet A. 2014;164A(11):2775-92. 558 30. Filges I, Hall JG. Failure to identify antenatal multiple congenital contractures and fetal akinesia--559 proposal of guidelines to improve diagnosis. Prenat Diagn. 2013;33(1):61-74. 560 Ma L, Yu X. Arthrogryposis multiplex congenita: classification, diagnosis, perioperative care, and 31. 561 anesthesia. Front Med. 2017;11(1):48-52. 562 32. Tita AT, Andrews WW. Diagnosis and management of clinical chorioamnionitis. Clin Perinatol. 563 2010;37(2):339-54. 564 Rubia-Rubia J, Arias A, Sierra A, Aguirre-Jaime A. Measurement of body temperature in adult 33. patients: comparative study of accuracy, reliability and validity of different devices. Int J Nurs Stud. 565 566 2011;48(7):872-80. Kong X. Jiang L. Zhang B. Sun L. Liu K. Predicting chorioamnionitis in patients with preterm 567 34. 568 premature rupture of membranes using inflammatory indexes: a retrospective study. Taiwan J Obstet 569 Gynecol. 2023;62(1):112-8. 570 Sung JH, Choi SJ, Oh SY, Roh CR, Kim JH. Revisiting the diagnostic criteria of clinical 35. 571 chorioamnionitis in preterm birth. BJOG. 2017;124(5):775-83. Galletta MAK, Schultz R, Sartorelli MFGO, Guerra ECL, Agra IKR, Peres SV, et al. Clinical 572 36. 573 characteristics, complications, and predictive model of histological chorioamnionitis in women with

574 preterm premature rupture of membranes. PLoS One. 2023;18(4):e0283974.

575 37. Ridout AE, Horsley V, Seed PT, Simpson N, Tribe RM, Shennan A. The neutrophil-to-lymphocyte 576 ratio: A low-cost antenatal indicator of placental chorioamnionitis in women who deliver preterm 577 without clinical signs and symptoms of infection. Eur J Obstet Gynecol Reprod Biol. 2023;280:34-9. 578 Sukumaran S, Pereira V, Mallur S, Chandraharan E. Cardiotocograph (CTG) changes and maternal 38. 579 and neonatal outcomes in chorioamnionitis and/or funisitis confirmed on histopathology. Eur J Obstet 580 Gynecol Reprod Biol. 2021;260:183-8. 581 39. Afors K, Chandraharan E. Use of continuous electronic fetal monitoring in a preterm fetus: 582 clinical dilemmas and recommendations for practice. J Pregnancy. 2011;2011:848794. 583 40. Lee SM, Park KH, Jung EY, Kook SY, Park H, Jeon SJ. Inflammatory proteins in maternal plasma, 584 cervicovaginal and amniotic fluids as predictors of intra-amniotic infection in preterm premature 585 rupture of membranes. PLoS One. 2018;13(7):e0200311. Dorfeuille N, Morin V, Tétu A, Demers S, Laforest G, Gouin K, et al. Vaginal Fluid Inflammatory 586 41. 587 Biomarkers and the Risk of Adverse Neonatal Outcomes in Women with PPROM. Am J Perinatol. 588 2016:33(10):1003-7. 589 Lee SM, Park KH, Jung EY, Kook SY, Park H, Jeon SJ. Inflammatory proteins in maternal plasma, 42. 590 cervicovaginal and amniotic fluids as predictors of intra-amniotic infection in preterm premature 591 rupture of membranes. Plos One. 2018;13(7). [repeat of ref 40; to be addressed] 592 Kunze M, Klar M, Morfeld CA, Thorns B, Schild RL, Markfeld-Erol F, et al. Cytokines in 43. 593 noninvasively obtained amniotic fluid as predictors of fetal inflammatory response syndrome. Am J 594 Obstet Gynecol. 2016;215(1):96.e1-8. 595 44. Kacerovsky M, Musilova I, Bestvina T, Stepan M, Cobo T, Jacobsson B. Preterm Prelabor Rupture 596 of Membranes between 34 and 37 Weeks: A Point-of-Care Test of Vaginal Fluid Interleukin-6 597 Concentrations for a Noninvasive Detection of Intra-Amniotic Inflammation. Fetal Diagn Ther. 598 2018;43(3):175-83. Musilova I, Bestvina T, Hudeckova M, Michalec I, Cobo T, Jacobsson B, et al. Vaginal fluid IL-6 599 45 600 concentrations as a point-of-care test is of value in women with preterm PROM. Am J Obstet Gynecol. 601 2016;16:30438-0. 602 46. Kim SA, Park KH, Lee SM. Non-Invasive Prediction of Histologic Chorioamnionitis in Women with 603 Preterm Premature Rupture of Membranes. Yonsei Med J. 2016;57(2):461-8. 604 47. Hall M, Hutter J, Suff N, Avena Zampieri C, Tribe RM, Shennan A, et al. Antenatal diagnosis of 605 chorioamnionitis: A review of the potential role of fetal and placental imaging. Prenat Diagn. 606 2022;42(8):1049-58. 607 Aviram A, Quaglietta P, Warshafsky C, Zaltz A, Weiner E, Melamed N, et al. Utility of ultrasound 48. 608 assessment in management of pregnancies with preterm prelabor rupture of membranes. Ultrasound 609 Obstet Gynecol. 2020;55(6):806-14. 610 Caissutti C, Familiari A, Khalil A, Flacco ME, Manzoli L, Scambia G, et al. Small fetal thymus and 49. 611 adverse obstetrical outcome: a systematic review and a meta-analysis. Acta Obstet Gynecol Scand. 2018;97(2):111-21. 612 613 50. Story L, Zhang T, Uus A, Hutter J, Egloff A, Gibbons D, et al. Antenatal thymus volumes in fetuses that delivered <32 weeks' gestation: An MRI pilot study. Acta Obstet Gynecol Scand. 2021;100(6):1040-614 615 50. 616 51. Story L, Davidson A, Patkee P, Fleiss B, Kyriakopoulou V, Colford K, et al. Brain volumetry in 617 fetuses that deliver very preterm: An MRI pilot study. Neuroimage Clin. 2021;30:102650. 618 Hutter J, Slator PJ, Avena Zampieri C, Hall M, Rutherford M, Story L. Multi-modal MRI reveals 52. 619 changes in placental function following preterm premature rupture of membranes. Magn Reson Med.

620 2023;89(3):1151-9.

621 53. Tare M, Bensley JG, Moss TJ, Lingwood BE, Kim MY, Barton SK, et al. Exposure to intrauterine 622 inflammation leads to impaired function and altered structure in the preterm heart of fetal sheep. Clin 623 Sci (Lond). 2014;127(9):559-69. 54. Kuypers E, Wolfs TG, Collins JJ, Jellema RK, Newnham JP, Kemp MW, et al. Intraamniotic 624 625 lipopolysaccharide exposure changes cell populations and structure of the ovine fetal thymus. Reprod 626 Sci. 2013;20(8):946-56. 627 55. Kramer BW, Ladenburger A, Kunzmann S, Speer CP, Been JV, van Iwaarden JF, et al. Intravenous 628 lipopolysaccharide-induced pulmonary maturation and structural changes in fetal sheep. Am J Obstet 629 Gynecol. 2009;200(2):195.e1-10. Hoogenboom LA, Lely AT, Kemp MW, Saito M, Jobe AH, Wolfs TGAM, et al. Chorioamnionitis 630 56. 631 Causes Kidney Inflammation, Podocyte Damage, and Pro-fibrotic Changes in Fetal Lambs. Front Pediatr. 632 2022:10:796702 633 Suff N, Karda R, Diaz JA, Ng J, Baruteau J, Perocheau D, et al. Ascending Vaginal Infection Using 57. 634 Bioluminescent Bacteria Evokes Intrauterine Inflammation, Preterm Birth, and Neonatal Brain Injury in 635 Pregnant Mice. Am J Pathol. 2018;188(10):2164-76. Petit C, Deruelle P, Behal H, Rakza T, Balagny S, Subtil D, et al. Preterm premature rupture of 636 58. 637 membranes: Which criteria contraindicate home care management? Acta Obstet Gynecol Scand. 638 2018:97(12):1499-507. Kenyon S, Boulvain M, Neilson JP. Antibiotics for preterm rupture of membranes. Cochrane 639 59. 640 Database Syst Rev. 2010;4:CD001058. 641 60. Kenyon S, Taylor DJ, Tarnow-Mordi WO, ORACLE Collaborative Group. ORACLE antibiotics for 642 preterm prelabour rupture of the membranes: short-term and long-term outcomes. Acta Paediatr Suppl. 643 2002;91(437):12-5. Kenyon SL, Taylor DJ, Tarnow-Mordi W, ORACLE Collaborative Group. Broad-spectrum 644 61. 645 antibiotics for preterm, prelabour rupture of fetal membranes: the ORACLE I randomised trial. ORACLE 646 Collaborative Group. Lancet. 2001;357(9261):979-88. 62. 647 Carlo WA, McDonald SA, Fanaroff AA, Vohr BR, Stoll BJ, Ehrenkranz RA, et al. Association of 648 antenatal corticosteroids with mortality and neurodevelopmental outcomes among infants born at 22 to 649 25 weeks' gestation. JAMA. 2011;306(21):2348-58. 650 63. Althabe F, Belizán JM, McClure EM, Hemingway-Foday J, Berrueta M, Mazzoni A, et al. A 651 population-based, multifaceted strategy to implement antenatal corticosteroid treatment versus 652 standard care for the reduction of neonatal mortality due to preterm birth in low-income and middle-653 income countries: the ACT cluster-randomised trial. Lancet. 2015;385(9968):629-39. 654 64. McGoldrick E, Stewart F, Parker R, Dalziel SR. Antenatal corticosteroids for accelerating fetal 655 lung maturation for women at risk of preterm birth. Cochrane Database Syst Rev. 2020;12:CD004454. Martins I, Pereira I, Clode N. A pilot randomized controlled trial of complete bed rest versus 656 65. 657 activity restriction after preterm premature rupture of the membranes. Eur J Obstet Gynecol Reprod Biol. 2019;240:325-9. 658 659 66. Kovacevich GJ, Gaich SA, Lavin JP, Hopkins MP, Crane SS, Stewart J, et al. The prevalence of thromboembolic events among women with extended bed rest prescribed as part of the treatment for 660 661 premature labor or preterm premature rupture of membranes. Am J Obstet Gynecol. 2000;182(5):1089-662 92. 663 Royal College of Obstetricians and Gynaecologists. Report of the RCOG Working Party on 67. 664 prophylaxis against thromboembolism in Gynaecology and Obstetrics 1995. London: RCOG. 665 Nelson-Piercy C, MacCallum P, Mackillop L, on behalf of the Royal College of Obstetricians and 68. 666 Gynaecologists. Reducing the risk of venous thromboembolism during pregnancy and the puerperium.

667 2015; RCOG.

668 69. Shennan AH, Story L, on behalf of the Royal College of Obstetricians Gynaecologists. Cervical Cerclage. BJOG. 2022;129(7):1178-210. 669 670 70. Zullo F, Di Mascio D, Chauhan SP, Chrysostomou S, Suff N, Pecorini F, et al. Removal versus retention of cervical cerclage with preterm prelabor rupture of membranes: Systematic review and 671 672 meta-analysis. Eur J Obstet Gynecol Reprod Biol. 2023;288:83-9. 673 71. Wu J, Denoble AE, Kuller JA, Dotters-Katz SK. Management of Cerclage in Patients With Preterm 674 Prelabor Rupture of Membranes. Obstet Gynecol Surv. 2021;76(11):681-91. 675 Suff N, Kunitsyna M, Shennan A, Chandiramani M. Optimal timing of cervical cerclage removal 72. following preterm premature rupture of membranes; a retrospective analysis. Eur J Obstet Gynecol 676 Reprod Biol. 2021;259:75-80. 677 D Laskin M, Yinon Y, Whittle WL. Preterm premature rupture of membranes in the presence of 678 73. cerclage: is the risk for intra-uterine infection and adverse neonatal outcome increased? J Matern Fetal 679 680 Neonatal Med. 2012;25(4):424-8. 681 74. Hughes RG, Brocklehurst P, Steer PJ, Heath P, Stenson BM on behalf of the Royal College 682 ofObstetricians and Gynaecologists. Prevention of early-onset neonatal group B streptococcal disease. 683 Green-top GuidelineNo. 36. BJOG 2017;124:e280-e305. 684 Macfarlane PI, Wood S, Bennett J. Non-viable delivery at 20-23 weeks gestation: observations 75. 685 and signs of life after birth. Arch Dis Child Fetal Neonatal Ed. 2003;88(3):F199-202. 686 Hofmeyr GJ, Eke AC, Lawrie TA. Amnioinfusion for third trimester preterm premature rupture of 76. 687 membranes. Cochrane Database Syst Rev. 2014(3):CD000942. 688 77. Roberts D, Vause S, Martin W, Green P, Walkinshaw S, Bricker L, et al. Amnioinfusion in very 689 early preterm prelabor rupture of membranes (AMIPROM): pregnancy, neonatal and maternal 690 outcomes in a randomized controlled pilot study. Ultrasound Obstet Gynecol. 2014;43(5):490-9. 691 Clinical Trial: Treatment of Classic Mid-trimester PPROM by Means of Continuous Amnioinfusion 78. 692 (AmnionFlush). https://clinicaltrials.gov/ct2/show/NCT04696003. 693 Crowley AE, Grivell RM, Dodd JM. Sealing procedures for preterm prelabour rupture of 79. 694 membranes. Cochrane Database Syst Rev. 2016;7:CD010218. 695 80. Stramrood CA, Wessel I, Doornbos B, Aarnoudse JG, van den Berg PP, Schultz WC, et al. 696 Posttraumatic stress disorder following preeclampsia and PPROM: a prospective study with 15 months 697 follow-up. Reprod Sci. 2011;18(7):645-53. 698 Patel K, Williams S, Guirguis G, Gittens-Williams L, Apuzzio J. Genital tract GBS and rate of 81. 699 histologic chorioamnionitis in patients with preterm premature rupture of membrane. J Matern Fetal 700 Neonatal Med. 2018:31(19):2624-7. 701 Bond DM, Middleton P, Levett KM, van der Ham DP, Crowther CA, Buchanan SL, et al. Planned 82. 702 early birth versus expectant management for women with preterm prelabour rupture of membranes 703 prior to 37 weeks' gestation for improving pregnancy outcome. Cochrane Database Syst Rev. 704 2017;3:CD004735. 705 World Health Organisation. Clinical Practice Handbook for Safe Abortion. 2014; WHO. 83. 706 84. Russell P. Inflammatory lesions of the human placenta: Clinical significance of acute 707 chorioamnionitis. Placenta. 1980:3:227-44 708 85. Alfirevic Z, Milan SJ, Livio S. Caesarean section versus vaginal delivery for preterm birth in 709 singletons. Cochrane Database Syst Rev. 2012;6:CD000078. 710 Kayem G, Baumann R, Goffinet F, El Abiad S, Ville Y, Cabrol D, et al. Early preterm breech 86. 711 delivery: is a policy of planned vaginal delivery associated with increased risk of neonatal death? Am J 712 Obstet Gynecol. 2008;198(3):289.e1-6.

- 87. Impey LWM, Murphy DJ, Griffiths M, Penna LK on behalf of the Royal College of
- 714 Obstetriciansand Gynaecologists. Management of Breech Presentation. BJOG 2017; 124: e151–e177. 88.

Kawakita T, Reddy UM, Grantz KL, Landy HJ, Desale S, Iqbal SN. Maternal outcomes associated 715 716 with early preterm cesarean delivery. Am J Obstet Gynecol. 2017;216(3):312.e1-.e9. 717 89. Evans C, Goodings L, Hargitai B, Heazell A, Marton T, Miller N, et al. The Royal College of 718 Pathologists: Tissue pathway for histopatholigcal examination of the placenta. 2022; The Royal College of Pathologists. 719 Arora P, Bagga R, Kalra J, Kumar P, Radhika S, Gautam V. Mean gestation at delivery and 720 90. 721 histological chorioamnionitis correlates with early-onset neonatal sepsis following expectant 722 management in pPROM. J Obstet Gynaecol. 2015;35(3):235-40. 723 Kim CJ, Romero R, Chaemsaithong P, Chaiyasit N, Yoon BH, Kim YM. Acute chorioamnionitis and 91. 724 funisitis: definition, pathologic features, and clinical significance. Am J Obstet Gynecol. 2015;213(4):S29-725 52. 726 92. Armstrong-Wells J, Post MD, Donnelly M, Manco-Johnson MJ, Fisher BM, Winn VD. Patterns of 727 placental pathology in preterm premature rupture of membranes. J Dev Orig Health Dis. 2013;4(3):249-728 55. Nijkamp JW, Korteweg FJ, Holm JP, Timmer A, Erwich JJ, van Pampus MG. Subsequent 729 93. 730 pregnancy outcome after previous foetal death. Eur J Obstet Gynecol Reprod Biol. 2013;166(1):37-42. 731 Graham N, Stephens L, Johnstone ED, Heazell AEP. Can information regarding the index stillbirth 94. 732 determine risk of adverse outcome in a subsequent pregnancy? Findings from a single-center cohort 733 study. Acta Obstet Gynecol Scand. 2021;100(7):1326-35. Khong TY, Mooney EE, Ariel I, Balmus NC, Boyd TK, Brundler MA, et al. Sampling and Definitions 734 95. 735 of Placental Lesions: Amsterdam Placental Workshop Group Consensus Statement. Arch Pathol Lab 736 Med. 2016;140(7):698-713. Mangham LJ, Petrou S, Doyle LW, Draper ES, Marlow N. The cost of preterm birth throughout 737 96. 738 childhood in England and Wales. Pediatrics. 2009;123(2):e312-27. Newnham JP, Schilling C, Petrou S, Morris JM, Wallace EM, Brown K, et al. The health and 739 97

educational costs of preterm birth to 18 years of age in Australia. Aust N Z J Obstet Gynaecol.

741 2022;62(1):55-61.

742